Author's response to reviews

Title: Human Pregnane X receptor is expressed in human breast carcinomas, potential heterodimer formation between hPXR and RXR-alpha

Authors:

Isabel Conde (isabelcondemartin@yahoo.es)
Maria V T Lobo (mval.toledo@uah.es)
Javier Zamora (javier.zamora@hrc.es)
Julio Perez (juliop.marquez@uah.es)
Francisco J Gonzalez (figonzalez02@yahoo.es)
Emilio Alba (oncologia98@yahoo.com)
Benito Fraile (benito.fraile@uah.es)
Ricardo Paniagua (ricardo.paniagua@uah.es)
Maria I Arenas (misabel.arenas@uah.es)

Version: 6 Date: 18 June 2008

Author's response to reviews:

LETTER EXPLAINING THE FORMATING CHANGES MADE IN THE FINAL VERSION OF THE MANUSCRIPT

1. Title & Abstract. Please ensure these are identical in the manuscript and in the submission system.

We have checked the title and the abstract and now they are identical in the manuscript and in the submission system.

2. Affiliations - Please write each affiliation on a new line.

Each affiliations is on a new line now.

3. Each email address should be on a new line and should be preceded by the author's initials, not the full name.

This change has been done.